Author's response to reviews

Title: Trabeculectomy with versus without Releasable Sutures for Glaucoma: A Meta-Analysis of Randomized Controlled Trials

Authors:

Minwen Zhou (zmw8008@163.com)
Wei Wang (will.wong@qq.com)
Wenbin Huang (cyhwenb@gmail.com)
Xiulan Zhang (zhangxl2@mail.sysu.edu.cn)

Version: 6 Date: 10 March 2014

Author's response to reviews: see over
Dear Dr. Cruz,

Enclosed please find the revised version of our manuscript “Trabeculectomy with versus without Releasable Sutures for Glaucoma: A Meta-Analysis of Randomized Controlled Trials” (Manuscript ID: 1698011178110768).

We appreciate your supportive comments and the Reviewers’ thoughtful and constructive critiques, which had been extremely helpful in our revision. We have considered each of the reviewer’s comments, and have responded carefully and thoroughly as summarized in the following pages, which we believe that the article is thereby strengthened.

Thank you again for your further consideration of this manuscript.

Sincerely yours,
Xiulan Zhang, MD, PhD
Professor of Ophthalmology, Vice director of Glaucoma Department, Director of Clinical Research Center, Director of Institution of Drug Clinical Trials, Zhongshan Ophthalmic Center, State Key Laboratory of Ophthalmology, Sun Yat-sen University
54S.Xianlie Road, Guangzhou, China 510060
Email: zhangxl2@mail.sysu.edu.cn.
Tel: +86-20-87330484
Fax: +86-20-87333271;
1. We recommend that you copy edit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

Answer: Thank you. Following your suggestion, we have now improved the style of written English using an editing service of “www.edanzediting.com/bmc1”

2. Please also include the correction that you mentioned on the email before regarding the error of "the line 'page 3' appears within the text of the 'Background' section".

Answer: Thank you. We have corrected the “page 3” error.